\documentclass{article}
\usepackage{amsmath,amsfonts}
%\usepackage[OT4,OT1,T1]{fontenc}
%\usepackage[latin1,cp1255]{inputenc}
\usepackage{array}
\makeatletter
%\usepackage{babel}
%\newcommand{\theauthor}[1]{\hfill\begin{otherlanguage}{english}{\it #1}\end{otherlanguage} }
%\font\scf=ptmr8r scaled 910
%\font\kukua=ptmr8r scaled 1370
%\font\kukub=ptmr8r scaled 1200
%\usepackage{ij}
%\usepackage{times}
\newcommand{\rF}{\operatorname{F}}
\newcommand{\rG}{\operatorname{G}}
\newcommand{\rSp}{\operatorname{Sp}}
\newcommand{\bbR}{\mathbb{R}}

\textheight=550pt
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=9pt

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}


%\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
%{\halign {\normalbaselines%
%\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
%&\hbox to 1.3cm{\hfill ##}\cr%
%{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
%\def\indexauthor{}
%\def\endindexitem{}
%\def\indexpage{}
%\def\indextitle{}
\newcommand{\indexitem}[3]{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}

%\def\indexsee\indexauthor#1\partner#2\endindexitem
%{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\newcommand{\indexsee}[2]{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\makeatother

\begin{document}
\pagestyle{empty}

\index
\indexvolume 161
\indexyear 2007
\endindextitle

%3441 1
\indexitem{Acosta, M.\ D. and Sung Guen Kim}{Denseness of holomorphic functions attaining their numerical radii}{373}

%3427 2
\indexitem{Allen, P.}{Almost every 2-SAT function is unate}{311}

%3404 3
\indexitem{Anderson, G.~W.}{Digit patterns and the formal additive group}{125}

%3419 4
\indexitem{Aschenbrenner, M. and H.~Schoutens}{Lefschetz extensions, tight closure, and big Cohen-Macaulay algebras}{221}

%3434 5
\indexitem{Ceccherini-Silberstein, T. and M.~Coornaert}{Injective linear cellular automata and sofic groups}{1}

%3434 6
\indexsee{Coornaert, M.}{T.~Ceccherini-Silberstein}

%3420 7
\indexitem{Dilworth, S., E. Odell and B. Sari}{Lattice structures and spreading models}{387}

%3425 8
\indexitem{Fabi\'nska, E. and M.~Lassak}{Reduced bodies in normed planes}{75}

%3422 9
\indexitem{Greenberg, Matthew}{Lifting modular symbols of non-critical slope}{141}

%3421 10
\indexitem{Huang, Wen and Yingfei Yi}{A local variational principle and it applications to equilibrium states}{29}

%3426 11
\indexitem{Hug, D. and R. Schneider}{A stability result for a volume ratio}{209}

%3441 12
\indexsee{Kim, Sung Guen}{M.\ D.\ Acosta} 

%3408 13
\indexitem{Klep, I. and M.~Schweighofer}{A Nichtnegativstellensatz for polynomials in noncommuting variables}{17}

%3430 14
\indexitem{Koll\'ar, J.}{Algebraic varieties over PAC fields}{89}

%3417 15
\indexitem{Laskowski, M.~C.}{A simpler axiomatization of the Shelah-Spencer almost sure theories}{157}

%3425 16
\indexsee{Lassak, M.}{E.~Fabi\'nska}

%3412 17
\indexitem{Levcovitz , D., I. Vainsencher and F. Xavier}{Enumeration of cones over cubic scrolls}{103}

%3418 18
\indexitem{Narayanan, E.\ K.,  R.\ Rawat and S.\ K.\ Ray}{Approximation by $K$-finite functions in $L^p$ spaces}{187}

%3420 19
\indexsee{Odell, E.}{S.~Dilworth}

%3437 20
\indexitem{Urba\'nski, M. and A.\ Zdunik}{Instability of exponential Collet --- Eckmann maps}{347}

%3418 21
\indexsee{Rawat, R.}{E.~Narayanan}

%3418 22
\indexsee{Ray, S.~K.}{E.~Narayanan}

%3420 23
\indexsee{Sari, B.}{S.~Dilworth}

%3426 24
\indexsee{Schneider, R.}{D.~Hug}

%3419 25
\indexsee{Schoutens, H.}{M.~Aschenbrenner}

%3408 26
\indexsee{Schweighofer, M.}{I.~Klep}

%3412 27
\indexsee{Vainsencher, I.}{D.~Levcovitz}

%3412 28
\indexsee{Xavier, F.}{D.~Levcovitz}

%3421 29
\indexsee{Yi, Yingfei}{Wen Huang}

%3437 30
\indexsee{Zdunik, A.}{M.~Urba\'nski}

\end{document}